Author's response to reviews

Title: Determinants of the number of antenatal visits in a metropolitan region

Authors:

Katrien Beeckman (katrien.beeckman@vub.ac.be)
Fred Louckx (flouckx@vub.ac.be)
Koen Putman (kputman@vub.ac.be)

Version: 3 Date: 23 August 2010

Author's response to reviews:

Dear Madam,
Dear Sir,

We received the reviewers' comments on our manuscript: ‘Determinants of the number of antenatal visits in a metropolitan region.’ We are delighted that it is accepted for publication into your journal.

We prepared a final manuscript including the format changes that we were asked to make.

We hope that the present version can be accepted for publication in your journal.

Yours sincerely,
Katrien Beeckman